Dec. 18, 2021
Professor Dennis A Bloomfield,
World Journal of Clinical Cases Editor-in-Chief

Dear Professor Bloomfield:

I wish to re-submit the manuscript titled entitled “Ultrasound-guided rectus sheath block for anterior cutaneous nerve entrapment syndrome after laparoscopic surgery: A case report (NO:72470)”. The manuscript has been rechecked and the necessary changes have been made in accordance with the suggestions.

Thank you for your consideration. I look forward to hearing from you.

Sincerely,

Joho Tokumine, MD. PhD
Department of Anesthesiology, Kyorin University School of Medicine, 6-20-2 Sinkawa, Mitaka, Tokyo, 181-8611, Japan.
Telephone: +81-422-47-5511
Fax: +81-422-43-1504
E-mail: ii36469@wa2.so-net.ne.jp
Reviewer #1:
Scientific Quality: Grade B (Very good)
Language Quality: Grade A (Priority publishing)
Conclusion: Minor revision
Specific Comments to Authors: The conclusion should be more refined and highlight the core content of the article

Thank you for your comments. We revised the text.

Reviewer #2:
Scientific Quality: Grade B (Very good)
Language Quality: Grade A (Priority publishing)
Conclusion: Accept (General priority)
Specific Comments to Authors: There are more and more patients with anterior cutaneous nerve entrapment syndrome. Ultrasound-guided rectus sheath block is a new method with less trauma and good effect. This method is worth using for reference.

Thank you very much.